\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\usepackage{qrcode}

\def\abbrvJournalTitle{Int. J. of Clin. Pharm. Med. Sci.}
\def\jshort{ijcpms}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/ijcpms/}{\color{blue}\underline{\smash{https://pharmasprings.com/ijcpms/}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}
\usepackage{balance}

\makeatother


\def\citeauthor{~Pandala~Archana,
              ~Annabathula~Meghana,
              ~Hawashetty~Priyanka,
              ~Marella~Priyachandana,
              ~Allu~Sree~Reddy,
              ~Mogulla~Soujanya}
\def\citetitle{A Case Report on Lymphedema}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{float}

\begin{document}


\def\RunningAuthor{Pandala Archana et al.}
\firstPage{123}
\articleType{Case Report}
\receivedDate{25 Nov 2021}
\acceptedDate{12 Dec 2021}
\revisedDate{11 Dec 2021}
\journalVolume{1}
\journalIssue{4}
\copyrightYear{2021}
\eISSN{2583-0953}
\journalDoi{10.26452/ijcpms.v1i4.173}
\def\authorCount{6}
\def\affCount{1}

\def\journalTitle{International Journal of Clinical Pharmacokinetics and Medical Sciences}

\title{A Case Report on Lymphedema}
\author{Pandala~Archana\textsuperscript{*}\textsuperscript{\href{https://orcid.org/0000-0002-1290-087X}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Annabathula~Meghana\textsuperscript{\href{https://orcid.org/0000-0002-7651-3676}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Hawashetty~Priyanka\textsuperscript{\href{https://orcid.org/0000-0003-3768-1420}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Marella~Priyachandana\textsuperscript{\href{https://orcid.org/0000-0003-0259-0588}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Allu~Sree~Reddy\textsuperscript{\href{https://orcid.org/0000-0002-6696-5310}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}}, 
            Mogulla~Soujanya\textsuperscript{\href{https://orcid.org/0000-0001-8400-7864}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}}~\\[5pt]{St. Mary's Group of Institutions, Deshmukhi (Village), Pochampally (Mandal), Yadadri Bhuvanagiri (Dist)- 508284, Telangana, India}}

\begin{abstract}
Lymphedema can be a regarding approach about a disorder throughout the vascular system for which explanations femoral lump. Defective ingestion, as well as transportation containing lymph fluid via lymphatic vessels, cause accumulation of protein-wealthy liquid during interstitium, which results in traumatic inflammation of the overall femoral. Underived lymphedema will be regularly nominative at delivery. The rare instances that fact take the floor succeeding historic period 35 years will be terminated as lymphedema tarda. The amazing clients along with edema experience lump going from the decreased limbs top femoral edema will be a group a rare ailment. A 76-yr-old person was once provided along with type a 3-year record containing unilateral lump going from the leftover top limb.  Would have been no constitutional indicators as well as no test copy of lymphadenopathy or fundamental skin disorder. Blood exams, carcinoembryonic antigen viva, computer-assisted tomography scans, and venous Doppler ultrasound had also altogether ordinary. The overall trend analysis becomes primary higher femoral lymphedema. The lump that happens in the upper femoral lymphedema is often unchangeable and generally expands up to the hand. Almost 1/3 of those who are victims of the aliment herewith state seem to be treated as well as decreased limb lymphedema. Investigations are going to be allowed in incidents containing unilateral top limb lymphedema to prevent hidden malignancy and intrinsic complaint.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Lymphedema,\newline Upper Limb,\newline Systemic Ailment,\newline Swelling
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Pandala~Archana~\\ Phone:\ +91 6303015710~\\  Email:\ pandalaarchana1@gmail.com
        \par}\vspace*{-13pt}\hrulefill~\\{\fontsize{12}{14}\selectfont\ifx\@eISSN\@empty\else eISSN: \@eISSN\fi\ifx\@pISSN\@empty\else\hfill pISSN: \@pISSN\fi}\par% 
        \textsc{DOI:}\ \href{https://doi.org/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont\raggedright Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
Primary top limb lymphedema serves as a rare entity. Almost all reviews about lymphedema affecting the extremities confer with a disorder of one's decreased limbs \unskip~\cite{1295027:23889172}. Incidents will be most of the time categorized since number one or secondary. Primary lymphedema is antiquatedly depicted in particular within the smaller limbs \unskip~\cite{1295027:23889206}. Powerful majority containing instances consisting of underived lymphedema tend to be notion to be nonheritable(congenital). Heritage outmoded set up in several events, and also a gene outmoded recognized in some lives \unskip~\cite{1295027:23889208,1295027:23889209}. Secondary lymphedema serves as some usually identified; relevant instants incorporate affaire of one's limb succeeding inguinal inspection but without therapy the treatment consisting of breast carcinoma \unskip~\cite{1295027:23889211}. Immediate or occipital dysplasia of lymphatics or disorderly hyperplasia outmoded represented \unskip~\cite{1295027:23889213}. Other than primary lymphedema some normally seems to be primal successful lifestyles, critiques containing the biography undergo defined resulting displays of congenital edema \unskip~\cite{1295027:23889246}. Edema happening after age 35 years is now and then often known as lymphedema tarda.

A detailed medical assessment has to subtract secondary reasons for lymphedema. Lymphatic define in addition to lymphoscintigraphy is regularly finished to affirm the diagnosis however won't stand by essential inside the bulk containing incidents as soon as secondary reasons of lymphedema will be prohibited. The current document designates the general notecase containing a 76 year-2 months-old affected person along with unpredictable higher limb lymphedema \unskip~\cite{1295027:23889247}.

\textbf{Case Presentation}

A 76 year 2 months-vintage lady was provided to our health center with 4-11 months precedent days containing a lump of the proper higher femoral. There became nary related ache or intrinsical signs and symptoms. The affected person's precedent days become routine, and she or he stated nary suchlike family ancient times. Expect a mild lump of one's intact left top limb, findings in the investigation were ordinary [Figure~\ref{f-a38bd12a8c5e}]. She turned into in nary seeming distress. Her mucous had been red and wet, and her crucial symptoms had been regular. Work-study of the proper upper femoral (limb) discovered nontender, nonpitting edema. The temperature of a limb changed into ordinary, and there has been no state's evidence consisting of localized cellulitis.  Heart rates have been detectable. The apparent venous machine became every day. There were no bizarre conclusions along with a motion study of one's breasts and axillae and nary state's evidence containing generalized lymphadenopathy. A temporary diagnosis containing number one top limb edema became prosperous [Figure~\ref{f-8feba2f2e7ca}]. A full analysis containing blood assessments becomes finished, which includes hematology, electrolytes, liver function, and carcinoembryonic antigen exams. Chest x-rays, contrast computed tomography scans containing the chest and abdomen in each arterial and venous level, and an upper limb blood vessel (venous) Doppler ultrasound had been additionally completed. Entirely consequences have been every day. Mammography findings had been every day. Since the lump changed into symptomless, the overall patient plummeted connive at work operating room nauseant.


\bgroup
\fixFloatSize{images/0b21272e-da09-4a18-a917-6f9e364b14e8-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/0b21272e-da09-4a18-a917-6f9e364b14e8-upicture1.png}{\includegraphics{images/0b21272e-da09-4a18-a917-6f9e364b14e8-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Photograph of upper limbs showing swelling of the left side (dorsal aspect)}}
\label{f-a38bd12a8c5e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/e2e60c51-5ce8-4ce2-afcb-0fd63c81470c-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e2e60c51-5ce8-4ce2-afcb-0fd63c81470c-upicture2.png}{\includegraphics{images/e2e60c51-5ce8-4ce2-afcb-0fd63c81470c-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Photograph of upper limbs showing swelling of the left side (volar aspect)}}
\label{f-8feba2f2e7ca}
\end{figure}
\egroup

    
\section{DISCUSSION}
Lymphedema come out from the buildup containing protein-wealthy liquid inside the membrane areas, which explains due to a huge in spite of lymph that exceeds lymphatic machine truking talents. Ordinary drain consisting of lymph within the top limbs tends to occur thru thin-walled lymphatic terminals, which ever drainpipe to axillary nodes. Primary lymphedema could be due to aplasia or hypoplasia containing lymphatics or utilizing renal sickness.

Primary lymphedema outmoded hushed in the direction of through to three varieties on the overall prologuize containing age at surprise attack (onset). Lymphedema proposing at delivery may be referred that one may since congenital lymphedema (Milroy disease); from delivery as far as age 35 years, edema praecox (Meige disease); since age 35 years, lymphedema tarda. Primary lymphedema may have in addition also been related to whatever inherited syndromes and cutaneal problems, which come with pancake Turner syndrome, Klinefelter syndrome, and neurofibromatosis kind. Lymphedema outmoded reffered with in triplex websites inside the consistency (body), consisting containing the overall slighter limbs, torso, genital, head, and neck even participation going from intrinsic body parts were found out. Neverthless, preferred lymphedema does have in particular also been defined for arousing influence the decreased extremity.

Secondary lymphedema, that's almost commonplace inside the slighter limbs, consequences taken away lymphatic vessel obstruction resulting from a pathogenesis that includes maliciousness, burns, radioactivity, injury, or incision. Elephantiasis, a mosquito-borne contamination resulting from the parasite Wuchereria bancrofti, in particular, includes the slighter limbs and is the overall maximum commonplace form of secondary lymphedema international.

Studies containing primary lymphedema that included the higher limbs tend to be aspecial. All but situations containing higher limb lymphedema inside the biograghy comeout surgical lymph node incision for most cancers, malignant percolation, radioactivity, or injury. In girls, edema affiliated malignancy and is a common cause containing one sided limb lump. As a result of affiliation, malignancy has to be taken into consideration in all instances of one sided femoral lump. Ephithelial Ovarian cancer and Endometrial cancer have been most not unusual malignancy.

An x-ray of the lymphatic system has superseded lymphography as option in reference to confirmative the diagnosis containing lymphedema. CAT scanner and nuclear magnetic resonance imaging were going to be really useful in indicating the nodule structure and have the additional advantage in reference to capable of leaving aside fundamental ailment. In case, an evaluation computed tomography scan turned into done along with prohibited the general presence of systemic disorder.

Such a big amount of techniques as inhabiting lymphedema be mentioned in the literature. Many of those methods(strategies) will be comfirmative in place going from peace offering a specific treatment, further more the result have always been outcomes normally unpredictable. Conservative treatment plans have been tested, and variable residue, further more consist in reference to elevation, sequential compression device, gradient compression stockings, pumps, physical treatment, and rub down. Special strategies known as ``whole decongestive physiotherapy'' undergo additionally had encouraging results. The one in question technique may have fought a reputation in Europe Union as well as comprises multilayer compression bandaging, rubdown, and lengthy-term apply going from compressing garments as well as graded physical activities. Solid result are mentioned long time the one in question approach is administered by way of professional person practitioners.
    
\section{CONCLUSION}
Primary lymphedema of one's bigger femoral (higher limb) could be a rare phenomenon. Nearly all of circumstances will be unilateral, and lump (swelling) is often firm further more involves the general handwriting. Most 1/3 going from those who are victims of the aliment are going to be along with primary lymphedema attending and participation going from both upper along with decreased limbs. Serious research can be authorized in situations of individual higher limb lymphedema that one may prevent occult malignity and fundamental ailment.
    
\section{ACKNOWLEDGEMENT\textbf{\space }}
I would like to thanks Principal sir (Dr Kamal Hassan) St. Mary's Pharmacy College, Deshmukhi (Village), Pochampally (Mandal), Yadadri Bhuvanagiri (Dist)-508284, Telangana, India. 

\textbf{Conflict of Interest} \textbf{\space }

The authors attest that they have no conflict of interest in this study. 

\textbf{Funding Support} \textbf{\space }

The authors declare that there is no financial support for the current study.
    
\balance 

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
